Author's response to reviews

Title: Cystatin E/M suppresses legumain activity and invasion of human melanoma

Authors:

Jon J Briggs JJB (jbriggs33@yahoo.com)
Mads H Haugen MHH (Mads.Haugen@rr-research.no)
Harald T Johansen HTJ (h.t.johansen@farmasi.uio.no)
Adam I Riker AIR (Ariker@ochsner.org)
Magnus Abrahamson MA (Magnus.Abrahamson@med.lu.se)
Øystein Fodstad ØF (oystein.fodstad@medisin.uio.no)
Gunhild M Mælandsmo GMM (Gunhild.Mari.Malandsmo@rr-research.no)
Rigmor Solberg RS (rigmor.solberg@farmasi.uio.no)

Version: 4 Date: 12 January 2010

Author's response to reviews: see over
Dear Editor,

We were very pleased to know that our manuscript is acceptable for publication pending formatting requirements. Please find enclosed the final version of our manuscript with the formatting changes as requested. In addition, we have proof-read the manuscript and corrected some typing errors and performed some minor changes in the text.

Sincerely yours,
Rigmor Solberg